Author's response to reviews

Title: Effectiveness of a behavioural intervention to prevent excessive weight gain during infancy (The Baby Milk Trial): study protocol for a randomised controlled trial

Authors:

Rajalakshmi Lakshman Dr (rl284@mrc-epid.cam.ac.uk)
Fiona Whittle Ms (fiona.whittle@mrc-epid.cam.ac.uk)
Wendy Hardeman Dr (wh207@medschl.cam.ac.uk)
Marc Suhrcke Professor (marc.suhrcke@your.ac.uk)
Ed Wilson Dr (ed.wilson@medschl.cam.ac.uk)
Simon Griffin Professor (simon.griffin@mrc-epid.cam.ac.uk)
Ken Ong Dr (ken.ong@mrc-epid.cam.ac.uk)

Version: 4
Date: 2 September 2015

Author's response to reviews:

Dear Jessie
I have made the following changes to the manuscript
1. Included the date of registration with the trial registration number at the end of the Abstract.
2. Removed the word counts from the title page.
3. After the Discussion, included a statement- Trial status: Recruiting at time of submission.

Many thanks for your consideration.
Kind regards
Dr R Lakshman